Author's Response To Reviewer Comments

Dear Editor and Reviewers,

Thanks again for the constructive comments and the approval of our study. We have now include a paragraph in the discussion section describing the difference in off-target effects between Cas9 and dCas9 methyltransferases and effector proteins.

Kr Alun